Author's response to reviews

Title: Analysis of single-nucleotide polymorphisms (SNPs) in human CYP3A4 and CYP3A5 genes: potential implications for the metabolism of HIV drugs

Authors:

Giulia Berno (giulia.berno@inmi.it)
Mauro Zaccarelli (mauro.zaccarelli@inmi.it)
Caterina Gori (caterina.gori@inmi.it)
Massimo Tempestilli (massimo.tempestilli@inmi.it)
Andrea Antinori (andrea.antinori@inmi.it)
Carlo Federico Perno (cf.perno@uniroma2.it)
Leopoldo Paolo Pucillo (leopoldo.pucillo@inmi.it)
Roberta D'Arrigo (roberta.darrigo@inmi.it)

Version: 3
Date: 21 June 2014

Author's response to reviews: see over
COVER LETTER FOR SUBMISSION OF MANUSCRIPT AFTER REVIEWER’S REPORT

I am enclosing herewith a manuscript entitled “Analysis of single-nucleotide polymorphisms (SNPs) in human CYP3A4 and CYP3A5 genes: potential implications for the metabolism of HIV drugs” for publication in BMC (technical advance) for possible evaluation.

My comments after Editors' evaluation

The term “full-length” has been removed in the following pages:

Abstract, page 2, line 12.
Background, page 3, line 26.
Results, page 6, line 1.
Results, page 7, line 25.
Results, page 7, line 32.
Results, page 7, line 34.
Discussion, page 8, line 10.
Conclusions, page 9, line 1.

- All authors of this research paper have directly participated in the planning, execution, or analysis of this study;
- All authors of this paper have read and approved the final version submitted;
- The contents of this manuscript have not been copyrighted or published previously;
- The contents of this manuscript are not now under consideration for publication elsewhere;
- The contents of this manuscript will not be copyrighted, submitted, or published elsewhere, while acceptance by the Journal is under consideration.
Submitted manuscript is a Research Article.

This research project was conducted from __2012/01/01____ and it is on going.

My Research Project is fully sponsored by Italian Ministry of Health with grant number 262/RF-2009- 1547039 and I’m principal Investigator of the project.

Date: 06/20/2014

Roberta D’Arrigo, corresponding author